Author’s response to reviews

Title: Adherence to long-term use of renin-angiotensin II-aldosterone system inhibitors in children with chronic kidney disease

Authors:
Chien-Ning Hsu (chien_ning_hsu@hotmail.com)
Shiou-Huei Huang (cnhsu10@gmail.com)
You-Lin Tain (tainyl@cgmh.org.tw)

Version: 2 Date: 08 Feb 2019

Author’s response to reviews:

Dear Editor,

We appreciate editor's comments and reminders. The following comments have been addressed in the revision.

1. Please include the institutional email addresses of all authors on the title page of your manuscript.

==The institutional email addresses of three authors have been added on the title page.

2. Please provide a “List of abbreviations” subsection after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

==The abbreviations have been into page 14.

3. Please add a heading "Declarations" after your List of abbreviations before the References section. Under this heading be sure to add the following sub-sections:

== The Declarations title and the sub titles have been added into page 14-15.
4. As indicated in our submission guidelines, please include the following sections in your manuscript: Background, Methods, Results, Discussion, Conclusions.

== The section titles all have been revised as indicated.

5. Currently, the listed corresponding author in our submission system and your title page is different. Please take this opportunity to revise either so that the correct corresponding author is the same in both locations.

== Thank you. Dr. Tain is indexed as the corresponding author.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

== The clean version of revision has been made.